Author's response to reviews

Title: Epidural lipomatosis and congenital small spinal canal in spinal anesthesia: a case report

Authors:

Per P Flisberg (per.flisberg@skane.se)
Bo B Geijer (bo.geijer@lthalland.se)
Ulf U Schott (ulf.schott@skane.se)

Version: 2 Date: 4 March 2008

Author's response to reviews:

Change of age - patient age 26 not 35 years